DNB Working Paper

No. 541 / December 2016

Hazardous tango: Sovereign-bank interdependencies across countries and time

Jack Bekooij, Jon Frost, Remco van der Molen and Krzysztof Muzalewski

DeNederlandscheBank

EUROSYSTEEM

Hazardous tango: Sovereign-bank interdependencies across countries and time

Jack Bekooij, Jon Frost, Remco van der Molen and Krzysztof Muzalewski *

* Views expressed are those of the authors and do not necessarily reflect official positions of De Nederlandsche Bank.

Working Paper No. 541

De Nederlandsche Bank NV P.O. Box 98 1000 AB AMSTERDAM The Netherlands

December 2016

Hazardous tango: Sovereign-bank interdependencies across countries and time*

Jack Bekooij^a, Jon Frost^{a,b}, Remco van der Molen^a and Krzysztof Muzalewski^c

^a De Nederlandsche Bank, Amsterdam, The Netherlands ^b VU University, Amsterdam, The Netherlands ^c Unibail-Rodamco, Schiphol, The Netherlands

23 December 2016

Abstract

Sovereign-bank feedback loops have been at the heart of the euro area crisis and many previous debt crises. We regress a market measure of interdependency – the correlation between sovereign and bank credit default swaps (CDS) – against various fundamental indicators of interlinkages and risk for 65 banks from 23 countries from Q1 2006 to Q4 2015. We find evidence that direct sovereign debt holdings of banks, implicit contingent liabilities of the government to banks and market volatility are significantly linked to higher correlations. While such CDS correlations are generally higher for banks in countries bank-based financial systems, we do not find these channels to be stronger in these countries than market-based systems. Finally, we find that bank CDS levels perform better in explaining sovereign CDS levels in periods of high volatility. Overall, these results support the notion of non-linear effects and spillovers in CDS markets.

Keywords: sovereign debt, banking crises, financial stability, CDS markets. **JEL classifications:** G01, G21, H63.

^{*} With thanks to Patty Duijm, Cees Ullersma, Clemens Bonner, Robert Vermeulen, Ewout van Ommen, Jeroen Hessel, Jan Kakes, Paul Hilbers, Jakob de Haan (all DNB), Bo Becker (Stockholm School of Economics) and participants at a DNB lunch seminar for comments.

1. Introduction

The outbreak of the European sovereign debt crisis in 2010 underscored the danger of interdependencies between sovereigns and the banking sector. In Ireland, stress in the banking sector led the government to provide large-scale support in the form of capital injections and guarantees – a "pyrrhic victory" which ultimately compelled the government to request external EU/IMF support (Acharya, Drechsler and Schnabl, 2014). The fiscal stress and ultimate restructuring of the Greek government, meanwhile, led to the insolvency and restructuring of the Greek and Cypriot banking sectors (Zettelmayer, Trebesch and Gulati, 2013).

Since then, the so-called "doom loop" (Gros, 2013), "diabolic loop" (Cooper and Nikolov, 2015), "deadly embrace" (Farhi and Tirole, 2015) or "hazardous tango" (Merler and Pisani-Ferry, 2012; DNB, 2015) between banks and sovereigns has become the topic of intensive policy and academic discussions.¹ The research of Bolton and Jeanne (2011), Blundell-Wignall and Slovik (2011) and Alter and Schüler (2012) underscores the relevance of these interdependencies in the recent global financial crisis. Yet such links have also played a role in a number of earlier advanced economy and emerging market crises, from the sovereign defaults and banking crises in Denmark in 1813 to Russia in 1998 and Argentina in 2002 (Reinhart and Rogoff, 2009; 2011).

Despite these contributions, Gray and Jobst (2013) point out that the ultimate impact of such interdependencies on the economy is not entirely understood. Specifically, the impact may be different during normal times than during periods of stress. There may also be important differences between countries based on their market structures. As long as these issues remain unclear, credit risk and pro-cyclical tail risks may remain mispriced and there may be volatile swings in perceptions of the debt sustainability of sovereigns and the banking sector (De Grauwe and Ji, 2013). Only once the relationship between sovereign and banking risk is better understood can effective policies and regulatory measures be implemented to reduce the probability and impact of financial crises. Consequently, in-depth analysis of sovereign-bank interdependencies is essential from a supervisory, financial stability and monetary policy perspective.

This paper investigates which factors drive interdependencies between the credit risk of sovereigns and domestic banks across countries. To tackle this question, we draw on existing theory and empirical work on sovereign risk, banking crises and credit market frictions to distinguish between four potential channels of risk transfer: (i) direct bank-to-sovereign risk transfer; (ii) direct sovereign-to-bank risk transfer; (iii) indirect links through the real economy; and (iv) credit market frictions. The first three channels can be considered fundamental factors of banking and sovereign risk, while the fourth can be considered the result of market inefficiencies.

Our empirical analysis focuses on the level of sovereign-bank interdependencies and the drivers

¹ Throughout the paper, we use "interdependency" to mean a bidirectional relationship between the risk profile of a government and of domestically owned banking groups. A "feedback loop" is a special case of such interdependencies, when risk factors for either banks or sovereigns lead to a self-reinforcing deterioration of credit risk.

of these links over a broad sample of countries and over the past decade. To assess interdependency, a market measure is applied: the Spearman rank correlation between credit risk returns as given by daily changes in credit default swap (CDS) spreads of individual banks and their home sovereign. This correlation measures whether the risk premiums of banks and sovereigns hit extreme levels simultaneously. As an identification strategy, we analyze time series changes and cross-sectional differences in this correlation measure against the independent variables intended to proxy the four channels sketched above. We complement this with estimations of the *level* of CDS spreads of banks and sovereigns. These regressions are estimated for 65 banks in 23 countries with quarterly data over the period Q1 2006-Q4 2015. Our multivariate panel regressions make it possible to determine which factors play the most important role in sovereign-bank interdependencies, based on their statistical and economic significance. The innovation of our analysis is to assess these links not only for a specific stress period, but for a broad range of countries and banks over the period before and after the global financial crisis.

The results of these exercises show that both the exposures by a bank to sovereign debt and the contingent liabilities of the sovereign toward banks are positively and significantly linked to higher correlations between bank and sovereign CDS. While such CDS correlations are generally higher for banks in bank-based financial systems, we do not find these channels to be stronger in these countries than market-based systems. Among control variables, we find that sovereign-bank interdependency is higher in periods of low output growth and greater financial market volatility. Based on a second set of regressions, we find that sovereign CDS levels are explained by (lagged) bank CDS levels to a greater extent in periods of high volatility. Bank CDS levels show a strong link with lagged sovereign CDS levels across all periods, with no significant additional effect when volatility is high. From these results, we infer that sovereign and bank risks may spill over to one another through direct financial links, both in bank-based systems and market-based financial systems, and that bank-to-sovereign risk transfer is especially important during high volatility in financial markets.

This paper is structured as follows. Section 2 describes the emerging theory of sovereign-bank interlinkages based on the relevant literature, and derives testable hypotheses. Section 3 introduces the data, our estimation methodology and some broad trends in the data. Section 4 provides the empirical results in terms of interdependency analysis, regression results and the robustness checks. Finally, section 5 concludes.

2. Literature and hypotheses on sovereign-bank interlinkages

In order to structure the growing literature on sovereign-bank interdependencies, this section divides the theoretical models based on the channels that they emphasize. Where relevant, examples from the empirical literature are also given. It concludes by deriving testable hypotheses.

Figure 1 gives a schematic overview on the different channels through which sovereigns and

banks are interlinked, and shows how these different channels may result in feedback-loops between sovereigns and banks. In general, we can distinguish between: (i) direct bank-to-sovereign risk transfer, such as explicit and implicit government guarantees; (ii) direct sovereign-to-bank risk transfer, such as direct holdings of sovereign debt, correlations in the cost and availability of funding, and the credibility of government support to banks; (iii) indirect links, for example through credit risk in the real economy; and (iv) credit market frictions, which can lead prices of credit risk for both banks and sovereigns to persistently diverge from fundamental values.²



Figure 1 – Transmission channels of sovereign and bank risks

Note: solid arrows refer to direct transmission channels, while dashed arrows are indirect links. Red denotes fundamental factors, while the blue channel is the result of credit market frictions.

The size and relative importance of these channels may differ significantly across countries and over time. For example, we expect differences between countries where credit to the private sector is dominated by bank lending ("bank-based financial systems"), and where non-bank players and capital markets play a larger role ("market-based financial systems;" see Goldsmith, 1969; Levine, 2002; Gambacorta et al., 2014). In bank-based systems, the government may be more likely to bail out banks, given their importance in financing of the real economy, and may also depend to a greater extent on banks as investors in domestic sovereign debt markets. Moreover, there may be important differences between normal times and more volatile periods, such as financial crises. In normal times, the interdependencies are not problematic: the government can act as a safety net for individual banks, and government debt plays an important role as a safe asset. In periods of high

² For further description, see BIS (2011), ESRB (2015), Gray, Merton and Bodie (2008) and Acharya, Drechsler and Schnabl (2014). For a theoretical model on the interplay of bank equity and resolution, see Cooper and Nikolov (2015).

volatility, the interdependencies may become more problematic, as the different channels may result in a vicious circle, feedback loops or non-linear "cliff effects." These ultimately lead to higher financing costs and a greater risk of systemic crises. Particularly problematic are so-called "twin crises" with defaults by the sovereign and the banking sector, which may be larger and more costly than independent crisis events (see Panizza and Borenzstein, 2009; Reinhart and Rogoff, 2011; and Balteanu and Erce, 2014). Finally, when there is strong financial integration, for example in the euro area, there may be important cross-border effects of sovereign debt holdings by foreign banks (Bolton and Jeanne, 2011).³ Due to data limitations, we will only focus on domestic effects and cross-border contagion will be outside the scope of our analysis.

2.1 Direct bank-to-sovereign links

The experience of financial crises over the past decades and the large-scale support provided to financial institutions has given impetus to work on the effects of state support on the affected banks and sovereigns. In these cases, both explicit support to banks (direct capital and liquidity support, deposit insurance) and implicit guarantees (expectations of bail-out) can improve bank solvency, but may erode the solvency of the government. A key study is Gray, Merton and Bodie (2008) who show with contingent claims analysis (CCA) that the interdependencies are determined by the amount of *explicit* and *implicit* guarantees of the sovereign to banks. Acharya, Drechsler and Schnabl (2014) develop a theoretical model which describes the sovereign-bank relationship. Similarly to Gray, Merton and Bodie, they find that large support packages provided by sovereigns to the banking sector increase the degree of spillovers from banks to sovereigns.

Meanwhile, Bénassy-Quéré and Roussellet (2014) develop a micro-based measure of implicit contingent liabilities of the sovereign to the banking sector. This is calculated as the difference between the risk-weighted assets (RWA) and the core Tier I capital held by banks, i.e. the amount of risky assets not covered by a bank's own funds. A proxy of this measure, when consistent data on core Tier I capital is not available, is the RWA minus total equity. This can in turn be scaled relative to GDP, which approximates the government's carrying capacity; this measure will be used in our empirical estimations. A large literature including Panetta et al. (2009) and King (2009) examines the impact of state support on banks, and generally finds that interventions lower credit spreads for supported institutions. Yet they do not provide an analysis of the effects of support on the sovereign or on the degree of sovereign-bank interdependencies.

2.2. Direct sovereign-to-bank links

The most straightforward channel through which banks can be affected by sovereign risk is

³ Relatedly, Breckenfelder and Schwaab (2015) show that in the euro area, expectations of state support can extend across borders, as new information on banks from the ECB's 2014 Comprehensive Assessment affected not only the governments of stressed banks, but also non-stressed governments. Mink and De Haan (2013) show that news about Greece in 2010 had little impact on the equity prices of other European banks, while news about a potential bail-out had a large impact, perhaps due to its signal about future bank bailouts.

through direct holdings of sovereign debt. When government bonds make up a significant portion of banks' assets, changes in the price of sovereign securities – or sovereign default – directly affect bank balance sheets. Losses may cause a decrease in bank's profits, consequently decreasing equity and leading to investor concerns about the solvency of the bank. Moreover, the use of government bonds for liquidity purposes, e.g. as high-quality collateral in private transactions and in central bank liquidity operations, will be impaired by changes in sovereign risk measures.

Gennaioli, Martin and Rossi (2014) model such holdings and the risk of government default. They show that large-scale bank holdings of sovereign debt may be a key reason why advanced economy sovereigns do not default more often – but also why defaults are extremely costly when they do occur. Their theoretical predictions are that sovereign defaults should cause a contraction in private credit, and that this effect should be larger in countries where financial institutions are more developed and banks hold more government bonds. This is confirmed empirically with a panel of emerging and developing countries over 1980-2005. Similarly, D'Erasmo and Mendoza (2016) show theoretically that governments may be able to sustain higher levels of domestic public debt when they have a bias in favor of domestic bondholders.

Among applied empirical work, Blundell-Wignall and Slovik (2011) show that the scale of (domestic) sovereign exposures by Greek and Cypriot banks, as well as some Northern European banks (e.g. Hypo Real Estate) in early 2010 was so large relative to Tier I capital that these banks would not be able to absorb various potential haircuts to the value of such debt. Asonuma, Bakhache and Hesse (2015) show from a large panel of advanced and emerging market economies that greater home bias in the sovereign debt holdings of domestic banks can reduce the cost of borrowing for the sovereign, but can also allow governments to sustain higher deficits and to delay fiscal consolidation until public debt reaches dangerously high levels. Balteanu and Erce (2012) show that, as compared to isolated banking crises, "twin" bank and sovereign debt crises tend to feature a high level and strong growth of exposures by the banking sector to the sovereign.

Banks may increase government bond holdings during a crisis for a variety of reasons. Acharya and Steffen (2015) argue that purchases by euro area banks of peripheral bonds during 2007-2012 can be understood as "carry trade" behavior, or an attempt by "large, under-capitalized banks to exploit government guarantees, arbitrage regulatory risk weights, and access central-bank funding." Ongena, Popov and van Horen (2016) show that banks in stressed euro area countries tended to purchase more bonds than foreign banks in months of strong issuance, particularly when they had received state support – a finding linked by the authors to moral suasion. Alternatively, Castro and Mencía (2014) relate such increases to macroeconomic factors, such as industrial production and unemployment, which may indicate a lack of private credit demand.

In addition to the risks to banks after a sovereign default, banks' sovereign exposures may crowd out private credit, through various channels. Bottero, Lenzu and Mezzanotti (2015) find that the Greek bailout in 2010, which led to a reassessment of the riskiness of sovereign exposures of

banks, also led to a tightening in the credit supply of banks to firms. They estimate that the drop in bank lending by Italian banks to firms reached 2 percent over the subsequent year. Similarly, Popov and van Horen (2012) find significant effects of sovereign debt degradation on the real economy via the bank lending channel and the quality of the banking sector.

2.3. Indirect links

Sovereigns and banks can also be interdependent through their joint reliance on the real economy. Theoretically, Di Iasio and Pierobon (2013) show how the creditworthiness of the sovereign can affect bail-out expectations and liquidity risk of private intermediaries, including shadow banks. When creditworthiness deteriorates, this could force deleveraging pressures through financial markets, which can have significant effects on the real economy. Empirically, Angelini, Granda and Panetta (2014) find that the correlations between sovereign and bank CDS are not higher than correlations between sovereigns and non-financial companies in the euro area. They interpret this as evidence that "country risk seems to be a key factor underlying the sovereign-bank relationship." This issue will be returned to in an extension to our baseline model in section 4.3, where we examine differences between the sovereign-bank and "sovereign-corporate" nexus.

2.4. Credit market frictions

Finally, both sovereign and bank risk may be affected by frictions in credit markets, e.g. due to information asymmetries or to coordination failures, particularly during periods of stress. In this vein, Broner, Erce, Martin and Ventura (2014) develop a theoretical model in which domestic and foreign investors can trade sovereign debt in the secondary market, but where creditor discrimination leads to a higher expected return for domestic investors when default risk rises. At the same time, due to financial frictions, such purchases crowd out private borrowing, thus displacing productive investment and reducing growth and welfare.

There is a large empirical literature seeking to explain movements in credit market spreads (see for example, Collin-Dufresne, Goldstein and Martin, 2001 and Haugh, Ollivaud and Turner, 2009). For the euro area crisis, De Grauwe and Ji (2013) argue that the lack of explanatory power of debtto-GDP ratios before the crisis, and the dramatic (non-linear) increase in spreads since 2008, is evidence of such mispricing in the euro area CDS market. Yet as de Haan, van den End and Hessel (2013) show, estimations of credit spreads may be heavily influenced by modelling choices such as sample selection, inclusion of financial variables and whether coefficients are allowed to change across countries and over time. For this reason, such estimations must be approached with the proper care, and while some developments may appear to result from market frictions, it is difficult to isolate these definitively in practice.

2.5. Testable hypotheses

With these channels now described, we develop two testable hypotheses. Our first hypothesis exploits variation in our market measure of interdependency, namely Spearman correlations. One reason why the CDS premiums of banks and sovereigns would move together is because the assets and liabilities of domestic banks and the sovereign are directly linked, either through exposures of the banks to their sovereign, e.g. holdings of sovereign debt, or through (potential) commitments of the sovereign to domestic banks, e.g. explicit or implicit guarantees (sections 2.1 and 2.2 above). The higher these *direct* exposures, the more the riskiness of the sovereign and the banks will be related. An increase in the (perceived) riskiness of the one would lead to a higher (perceived) riskiness of the other. Yet the impact of direct mutual exposures on the sovereign-bank interdependency may not be the same for all countries, and - in addition to indicators of the real economy (section 2.3) – may also depend on market structure. For example, in countries where the banking sector provides a larger share of overall credit and is thus more systemically important, the government is more likely to guarantee banking sector liabilities in stress periods. Similarly, when banks are more dominant in private credit provision, they may also be a more important segment of the investor base for government debt. Hence, we would expect to see higher sovereign-bank interdependencies in these countries than in countries where the banking sector is less dominant:

 H_1 : Higher exposures of banks to sovereigns and higher implicit contingent liabilities are tied to a higher market perception of sovereign-bank interdependencies (i.e. higher Spearman correlations), and these effects are larger in countries with bank-based financial systems.

Of course, it is possible that the direct exposure channels only affect short-term market correlations but not actual *levels* of risk premiums. Our second hypothesis seeks to exclude this possibility. Specifically, we investigate how CDS spreads of sovereigns spill over to banks and vice versa. We expect that the CDS premiums of banks will be higher when the sovereign is under strain, and that sovereign CDS rise when banks are in distress. Moreover, we would expect these spillovers to be especially large in periods of financial volatility. This yields our second hypothesis:

*H*₂: Sovereign credit spreads are a stronger determinant of bank credit spreads, and bank spreads are a stronger determinant of sovereign spreads, in periods of high market volatility.

This hypothesis can best be tested by estimating the level of sovereign and bank credit spreads directly. If the explanatory power of lagged sovereign (bank) credit spreads on bank (sovereign) spreads rises when volatility is high, this is further evidence of non-linear effects, such as a feedback loop. Moreover, because higher credit spreads for sovereigns lead to a greater fiscal burden, and higher spreads for banks tend to be passed on to private borrowers, examining the level of spreads

allows us to make a more direct link to potential welfare effects.

3. Data, methodology and trends

To investigate sovereign-bank interdependencies across countries and time, we introduce our data sources and our market measure of bank-sovereign links, namely the Spearman correlation between CDS premiums of a banking group and its home sovereign. Next, we empirically test which factors are the most important in explaining such links, by regressing our measure on specific bank, sovereign and country characteristics.

3.1. Data sources

The analysis on sovereign-bank interdependencies is performed on a panel data set of 65 banks in 23 countries at quarterly frequency over the course of 10 years (Q1 2006-Q4 2015). The country choice and the period for the analysis were restricted largely by data availability of the CDS series. For most countries and banks the series start in January 2006. The resulting data set spans the global pre-crisis (2006-2007), in-crisis (2007-2009) and recovery periods (2010-2015), and also the euro area sovereign debt crisis (2010-2012). For each of the 23 countries, daily CDS spreads of the sovereign and the largest banks by market capitalization were selected.⁴ The use of large banks is based on data availability, but this choice should provide a good basis for sovereign-bank interdependencies, since especially bank-to-sovereign risk transfer (expectations of state support) relates mostly to large, systemically important institutions. The list of selected countries and the included banks is presented in the annex.

The total assets of banking groups and their exposure to governments are taken from Bankscope. Because these variables are only available at annual frequency, they are interpolated within a year using quarterly data on the total claims and exposures of the full national banking sector from the IMF's International Financial Statistics (IFS). The underlying assumption is that quarterly changes in total assets and exposures by each individual bank are comparable with the changes of the aggregate national banking sector.⁵ Quarterly macroeconomic variables are taken from the IMF, and sovereign and bank ratings come from Moody's. For the level of CDS and the VIX, which are available on daily basis, quarterly averages are used. For the correlations, quarterly values are obtained by taking the correlation between changes in daily spreads within the quarter (see below).

⁴ For most countries, there are at least three banks in the sample. For some countries (Germany, Ireland, Switzerland and Turkey) only two banks were selected due to unavailability of CDS data or illiquidity of CDS series (e.g. for Deutsche Postbank and Anglo Irish Bank). For China, Denmark and Norway, CDS data were available for only one domestic bank.

⁵ The IFS data refer to the domestic banking sector on a locational basis, whereas the Bankscope banking group data are on a consolidated basis. The in-quarter adjustments are between -10% and 16% relative to a linear trend between annual values from Bankscope. While it is likely that individual banking groups diverge from the behavioral patterns of the entire banking sector within individual years, these differences are by definition short-term and in any case will not have been known publicly or by investors in bank CDS within individual years.

3.2. Measure of sovereign-bank interdependencies

Our measure for the degree of interdependencies between sovereigns and the domestic banking sectors is derived from the co-movement of the relevant CDS series. In particular, the CDS series of each bank is compared with its domestic sovereign's CDS series. In order to capture co-movement (i.e. the timing of large increases and decreases in credit risk, rather than the scale of such changes), the Spearman's rank correlation was employed. Since CDS spreads tend to be non-stationary, the degree of co-movement between returns instead of levels is measured. For each series, daily returns were calculated by taking differences between two end-of-the-day spreads.⁶ The returns in each series are assigned ranks according to their values. Once the ranking process is completed, the linear correlations between the rankings of sovereign and bank CDS are computed according to the standard Pearson's formula:

$$\rho_{CDS_B,CDS_S} = \frac{Cov_{(CDS_B,CDS_S)}}{\sigma_{CDS_B} * \sigma_{CDS_S}}$$

where $Cov_{(CDS_B,CDS_S)}$ is the covariance of the bank and sovereign CDS spreads and σ_{CDS_B} and σ_{CDS_S} are the standard deviation of each series. The resulting daily values are averaged out to produce quarterly data.⁷ Although we also use other correlation measures to test the robustness of our results, an advantage of the Spearman rank correlation is that it does not assume a linear relationship between the two series.

3.3. Explanatory variables

In order to determine the factors underlying the level of sovereign-bank interdependencies several measures representing the sovereign to bank and bank to sovereign exposures were developed. The preferred measures are the following two measures:

Sovereign exposures of banks – the claims on governments by a bank relative to its total assets. This series is on a consolidated basis. A key limitation is that it captures holdings of all government debt, rather than just the debt of the domestic sovereign. While domestic sovereign exposures are available for the entire banking sector in most EU countries (and are used in the robustness checks), they are not consistently available for individual banks, or for other regions. However, in most non-EU countries, domestic debt is a majority of banks' sovereign debt portfolios, making this a reasonable proxy.⁸ Larger holdings of

⁶ For example, the return on 1 May is the end of day price for 1May minus end of day price for the 30 April.

⁷ For alternative approaches, such as co-exceedance, see Baur and Schulze (2005) and Forbes and Rigobon (2002).

⁸ In the EU, domestic exposures make up 52% of overall government bonds holdings on a consolidated basis, and 75% on a locational basis. There are higher values in the larger countries, such as the UK, Spain, Italy, France and Germany. International supervisory data sources suggest that this share is much higher outside the EU.

sovereign debt should be associated with higher sovereign-bank interdependency.

Implicit contingent liabilities of the sovereign – in line with Bénassy-Quéré and Roussellet (2014), the amount of implicit liabilities is proxied by the RWA minus total equity of each bank, divided by the domestic country's GDP.⁹ Even though not contracted by law, there is an expectation of sovereign support to domestic banks in times of distress. Consequently, a larger and riskier banking sector with a lower equity cushion increases the amount of potential support. An increase in this measure should lead to increasing levels of sovereign-bank interdependencies.¹⁰

In addition to sovereign-bank exposures, bank and sovereign risk determinants are incorporated into the model. This relates to the notion that the state of the banking sector is an important determinant of sovereign CDS and vice versa (Dieckmann and Plank, 2011). As a result, sovereign and bank ratings can influence the level of sovereign-bank interdependencies. Furthermore, macroeconomic and financial market variables also partially account for the sovereign's and banking sector's common exposure to the same economy and the perception of risk. Consequently, two risk determinants, one macroeconomic variable and one financial market control were included in the baseline model. The risk determinants are:

- Sovereign Ratings sovereign credit ratings transformed into numbers with a linear numerical scale (AAA=25, intervals of 1 unit per notch). This variable represents an overall measure of sovereign creditworthiness. The sovereign credit rating is based on the Long Term Issuer rating of Moody's Investor Service. A higher rating represents greater sovereign creditworthiness, which may reduce sovereign-bank interdependencies.
- *Bank Ratings* the average of Moody's Long Term Deposits (Domestic currency) for each of the 65 banks, again transformed into numbers with a linear numerical scale (AAA=25, intervals of 1 unit per notch). Again, a higher rating represents a safer bank which poses less risk to the sovereign.

The macroeconomic and financial market controls are:

• *Output growth* – year-on-year change in real GDP, taken from the OECD. High economic growth generally entails an improving risk outlook and a greater capacity to pay down a nominal debt burden. This decreases the risk of future default and at the same time has a positive effect on overall creditworthiness of both banks and the sovereign.

⁹ Data sources: SNL Financial and IMF International Financial Statistics. An additional advantage of the Bénassy-Quéré measure over alternative indicators is that it does not include sovereign exposures, due to the zero risk weighting of sovereigns in the capital requirement framework in various global jurisdictions.

¹⁰ An avenue for future analysis would be to look at *explicit* contingent liabilities from banking sector support programs, and at any changes due to recent reforms to encourage bail-in.

- Volatility the Chicago Board Option Exchange volatility index ("VIX"), accessed from
 Datastream. The VIX is the most commonly used index of market volatility. Since it is a
 measure of global market volatility, the same index was used for every country in the
 sample. Daily index values were averaged over each quarter to match the rest of the sample.
 Controlling for volatility in the markets helps to isolate changes in CDS which are a result
 of common market movements (or the level of financial frictions) as opposed to a result of
 changes in fundamental variables.
- *Bank share of credit* the percentage share of banks in overall credit to the domestic private sector, taken from the BIS. This share is a relatively direct measure of financial structure, and allows for a classification of countries into relatively bank-based versus market-based financial systems.¹¹

3.4 Regression methodology

Our market-based measure of the perceived interdependency between the sovereign and banks is compared across countries and time, and regressed against relevant financial and macroeconomic variables. For our baseline regression, used to test the first hypothesis, this takes the form of:

$$\rho_{CDS_B,CDS_{S_{it}}} = \alpha + \beta_1 X_{i,t-1} + \beta_2 Y_{i,t-1} + \beta_3 Z_{i,t-1} + \beta_4 \sigma_t + \gamma_i [+\rho_{CDS_B,CDS_{S_{it}-1}}] + \epsilon_{i,t}$$
(1)

where $\rho_{CDS_B,CDS_{s_{i,t}}}$ denotes our correlation measure for bank *i* and quarter *t*, $X_{i,t-1}$ is a proxy of direct sovereign-to-bank risk transfer channels, lagged by one quarter; $Y_{i,t-1}$ is a proxy of direct bank-tosovereign risk transfer; $Z_{i,t-1}$ is a vector with bank and sovereign ratings and output growth; and $\sigma_{,t}$ is overall financial market volatility, proxied by the VIX, which is included contemporaneously (i.e. in the same quarter).¹² The variables α , γ , and ε are, respectively, the constant, the bank fixed effects term and an error term. The variables β_1 , β_2 , β_3 , and β_4 denote our estimated coefficients. These equations are estimated using ordinary least squares (OLS). A Hausman test confirms the choice of fixed effects, as random effects would be inconsistent. Due to some autocorrelation in the quarterly data, we also show the baseline regressions with the lagged dependent variable, $\rho_{CDS_B,CDS_{s_i,t-1}}$. Throughout the regressions, we use clustered standard errors.

To better test the differences between bank-based and market-based systems, we use simple interaction terms, such that the estimation becomes:

¹¹ Countries with relatively bank-based systems are those where the average share of bank credit in total credit is above the sample median, while market-based systems have bank credit shares below the median. Note that some traditionally bank-based countries, such as Germany and Japan, have seen a growing role of non-bank credit in the past decade, such that they are now relatively market-based. Emerging market economies generally have a large bank share of credit. Alternative data sources, such as the OECD, yield a similar relative ranking of countries, but do not have data on the bank share of credit for all countries over the full sample period.

¹² The independent variables are lagged by one quarter as a simple means of addressing reverse causality. The VIX is shorter-term and less likely to be influenced by individual countries in the sample.

$$\rho_{CDS_B,CDS_{s_{i,t}}} = \alpha + \beta_1 X_{i,t-1} + \beta_2 (X_{i,t-1} * S_{i,t-1}) + \beta_3 Y_{i,t-1} + \beta_4 (X_{i,t-1} * S_{i,t-1}) + \beta_5 S_{i,t-1} + \beta_6 Z_{i,t-1} + \beta_7 \sigma_t + \gamma_i + \epsilon_{i,t}$$
(2)

where $S_{i,t-1}$ is the financial structure of the country (i.e. the share of bank credit in overall credit), and $X_{i,t-1} * S_{i,t-1}$ and $Y_{i,t-1} * S_{i,t-1}$ are, respectively, the products of this measure and direct sovereign-to-bank and bank-to-sovereign risk transfer channels.

In line with our second hypothesis, we would like to exclude the possibility that correlations only concern short-term co-movement. Hence, in a second step, we estimate the levels of banks CDS and sovereign CDS spreads directly. These regressions take the form of:

$$CDS_{B_{i,t}} = \alpha + \beta_1 X_{i,t-1} + \beta_2 Y_{i,t-1} + \beta_3 Z_{i,t-1} + \beta_4 \sigma_t + \beta_5 CDS_{S_{i,t-1}} + \gamma_i + \epsilon_{i,t}$$
(3)

$$CDS_{S_{j,t}} = \alpha + \beta_1 \bar{X}_{j,t-1} + \beta_2 \bar{Y}_{j,t-1} + \beta_3 \bar{Z}_{j,t-1} + \beta_4 \sigma_t + \beta_5 CDS_{\bar{B}_{i,t-1}} + \gamma_j + \epsilon_{j,t}$$
(4)

where $CDS_{B_{i,t}}$ denotes the quarterly average of 5-year CDS spreads for bank *i* in quarter *t*, $CDS_{S_{j,t}}$ is the quarterly average 5-year CDS spreads of sovereign *j* in quarter *t*, and the other variables are the same as in equation (1). The key difference in equation (4) is that all variables are now at the country level, such that the sovereign-to-bank and bank-to-sovereign risk transfer indicators are a total for the entire banking sector (from IMF IFS). Meanwhile, $CDS_{\bar{B}_{i,t-1}}$ denotes an average for all of the banks from country *j* in the sample, and γ_j are now country fixed effects.

3.5. Descriptive statistics

Table 1 shows key characteristics of all of the variables used in the regressions. The correlation measure is usually positive, with a mean around 33%. Sovereign CDS spreads (5-year, expressed in basis points, bp) vary from a level close to 0 up to a level of 1294 bp for sovereigns (Portugal, Q1 2011) and 1698 bp for banks (Banco Commercial Portugues, Q4 2011). Sovereign and bank ratings have been converted to a number between 0 (lowest) and 25 (highest ratings). The implicit contingent liabilities average 77% of GDP, but with large heterogeneity between bank-country pairs. The maximum is 273% by Allied Irish Bank in Q1 2009. Bank exposures to sovereigns average at 8.2% of total assets, and are generally higher in emerging markets, e.g. the 44.7% recorded by Akbank in Q4 2009. The bank share of private credit averages at about 66% across all countries, but is much lower in countries like the US, Ireland and Luxembourg, and much higher in emerging markets like Brazil. Real GDP growth averages 3.4% across the sample, and the VIX averages at 20 bp, with a maximum level of 59 bp in Q4 2008, after the failure of Lehman Brothers.

Variable	Observations	Mean	Std. Dev.	Min	Max
Spearman correlation (in %)	2,040	33.2%	24.8%	-33.6%	93.7%
Sovereign CDS (in bp)	2,438	110.2	136.7	1.4	1294.7
Bank CDS (in bp)	2,442	168.9	188.2	3.6	1698.7
Implicit CL (RWA-TE, % of GDP)	2,225	77.4%	56.7%	1.4%	272.8%
Bank exposures to sovereigns (% total assets)	2,127	9.2%	7.6%	0.1%	44.7%
Bank share in private sector credit (BIS)	2,600	65.6%	17.3%	16.7%	95.3%
Sovereign rating (in notches)	2,600	21.62	4.03	13.00	25.00
Bank rating (in notches)	2,577	20.45	3.12	7.00	25.00
GDP growth (in %)	2,600	2.1%	3.4%	-13.1%	13.3%
VIX (in bp)	2,600	20.39	9.13	11.03	58.61

Table 1: Descri	ptive statistics of	lependent and ind	ependent variables
-----------------	---------------------	-------------------	--------------------

Note: unbalanced panel, as CDS data are missing for some countries in the early part of the sample period.

All variables have been tested for stationarity with the Levin-Lin-Chu (2003) panel unit root test. A unit root can be ruled out for all variables at the 95% confidence level, except for sovereign ratings. All variables have also been tested for multicollinearity (table 2). The correlation between the quarterly average levels of bank and sovereign CDS (80%) is much higher than the average of within-quarter Spearman correlations of daily changes, which are used in the regressions. The high correlation between sovereign and bank ratings (70%) calls for some caution in interpreting the effects of regressions that include both variables. The high (negative) correlation between sovereign ratings and sovereign CDS, and between bank ratings and bank CDS is in line with expectations, as more creditworthy banks and sovereigns have lower financing costs. In the second step, ratings will thus form an important control variable on the right-hand side of regressions on CDS spread levels.

	Spearman correlation	Sovereign CDS	Bank CDS	Implicit CL	Bank exp. to sovereigns	Bank share	Sovereign rating	Bank rating	GDP growth
Sovereign CDS	0.33								
Bank CDS	0.26	0.80							
Implicit contingent liabilities	-0.05	0.00	-0.04						
Bank exposures to sovereigns	0.06	0.12	0.13	-0.28					
Bank share in private credit	0.28	0.24	0.09	-0.20	0.03				
Sovereign rating	-0.31	-0.60	-0.48	0.32	-0.40	-0.54			
Bank rating	-0.17	-0.53	-0.53	0.26	-0.22	-0.31	0.72		
GDP growth	-0.09	-0.18	-0.23	-0.25	0.17	0.30	-0.29	-0.10	
VIX	0.19	0.19	0.21	0.10	-0.03	0.01	0.06	0.16	-0.37

Table 2: Correlations between the dependent and independent variables

Note: pairwise correlations, i.e. linear correlations considering all available observations per pair of variables.

3. 6. Trends in the data

Figure 2 shows the correlation of sovereign and bank CDS premiums over time, split by countries with relatively bank-based versus relatively market-based financial systems. The strong increase in the correlation measures between 2007 and 2009 is striking. This is unlikely to be the result of changes in fundamental factors alone, but could rather be interpreted as evidence of a repricing of risk through feedback loops. Moreover, we find that the sovereign-bank interdependency is generally higher for banks in countries with a bank-based financial system than in countries with a market-based financial system. Also, the decline in the correlation in the postcrisis period has been more pronounced in in countries with a market-based financial system.



Figure 2: Spearman rank correlations for different countries over time

Note: Countries with a ratio of bank credit to total credit above the sample-wide median as of Q3 2015 are defined as bank-based, while those with such ratios below the median are defined as market-based.



Figure 3: Spearman rank correlations for different countries before and after 2011

Note: In the left figure we show the movement of correlations in relation to the size of the banking sector in the period 2007Q1 to 2011Q3. In the right the same movement is presented for the period 2011Q3 to 2015Q4.

Figure 3 shows the change in banking sector size and correlation in sovereign and bank CDS throughout the sample period. Especially bank-based countries saw the size of the banking sector relative to GDP increase in the period through 2011, at the same time that the market measure of the sovereign-bank nexus for the largest banks was increasing. Relatively market-based countries saw the relative size of the banking sector decline, but correlations increase. Since the peak of the euro area crisis in 2011, the euro area countries (e.g. Italy and the Netherlands) have seen a decline in the overall size of the banking sector, and correlations have again decreased. Yet this is not exclusive to Europe: a similar pattern can be seen in the United States and Japan, which have also experienced bank deleveraging and a declining sovereign-bank nexus, and in the average of bankbased and market-based countries generally. Also note the different patterns for the United States and the Netherlands, on the one hand, and for Italy and Japan, on the other. The first two countries, where markets are sanguine about public debt sustainability, basically moved back to their old position after 2011. Italy and Japan, two countries with very high public debt levels, now have a bank-sovereign correlation substantially higher than before the crisis. This suggests that markets have paid more attention to the sovereign-bank nexus since the crisis.

4. Empirical results and robustness checks

4.1. Regression results

To test our <u>first hypothesis</u> we regress the correlation between bank and sovereign CDS premiums on banks' holdings of sovereign debt and the contingent liabilities of the sovereign to the bank. While our hypothesis focuses on the direct risk transfer channels (1 and 2 in figure 1), we also control for indirect links in the regressions. Common exposures to economic shocks can be captured with GDP growth, while financial shocks should be captured with the VIX. We also include bank fixed effects to take into account unobserved bank-specific characteristics.

The results are shown in table 3. For the first regression (column 1) with only the direct risk transfer channels and the VIX, we find that larger sovereign exposures are associated with a significantly higher correlation between the CDS premiums of banks and their sovereign. In economic terms, an increase in sovereign exposures by 4% of total assets (one standard deviation in Italy during the sample) is associated with a sovereign-bank CDS correlation that is 5.9 percentage points higher. Higher contingent liabilities, as measured by the Bénassy-Quéré and Roussellet (2014) measure, are also associated with a tighter sovereign-bank link. The economic significance is generally smaller: an increase in this measure by 7.5% of GDP (one standard deviation in Brazil or Turkey) is associated with an increase by 1.3 pp in the sovereign-bank CDS correlation. To test whether this effect depends on the riskiness of the sovereign holdings of banks, we include the sovereign rating and its interaction with sovereign exposures as explanatory variables (column 2). We find that these variables are not significantly related to the sovereign-bank CDS correlation. This suggests that higher sovereign exposures are associated with stronger

bank-sovereign interdependencies, irrespective of the riskiness of these exposures. Together, these results suggest that across all the banks and countries under analysis, both bank-to-sovereign links and sovereign-to-bank links are tied to higher interdependency. Moreover, these interdependencies are higher when financial market stress, as measured by the VIX, is high.

	(1)	(2)	(3) Full sample	(4)	(5)
Sovereign to Bank risk transfer					
Bank exposures to sovereigns (% total assets)	1.468***	1.439**	1.399***	0.604***	0.004
	(0.401)	(0.605)	(0.388)	(0.196)	(1.025)
Exposures to sovereigns * sovereign rating		-0.016 (0.082)			
Exposures to sovereigns * bank share					2.309
					(1.744)
Bank to Sovereign risk transfer					
Implicit CL (RWA-TE, % of GDP)	0.175***	0.174***	0.142**	0.083**	0.435***
	(0.061)	(0.059)	(0.066)	(0.036)	(0.149)
Implicit CL * bank share					-0.463* (0.250)
Risk measures					
Sovereign rating		-0.007	-0.002	-0.007**	
		(0.008)	(0.009)	(0.003)	
Bank rating			-0.001	0.006	
0			(0.010)	(0.004)	
Macroeconomic controls					
GDP growth			-0.010***	-0.002	
			(0.003)	(0.001)	
Bank share					0.03
Financial market volatility					(0.309)
VIX	0.006***	0.006***	0.006***	0.004***	0.006***
	(0.001)	(0.001)	(0.001)	(0.000)	(0.001)
Autocorrelation			. ,	. ,	
Spearman (lag1)				0.534***	
				(0.031)	
Number of observations	1602	1602	1598	1549	1602
Number of groups	65	65	65	64	65
R-sq within	0.103	0.105	0.117	0.401	0.114
R-sq between	0.008	0.000	0.010	0.700	0.010
R-sq overall	0.008	0.017	0.014	0.496	0.032

Table 3: Estimations of Spearman rank correlation between sovereign and bank CDS

Note: * significance at 90%; ** significance at 95%; *** significance at 99%. Bank fixed effects and clustered standard errors are used throughout. Constants are not reported.

When adding further control variables, i.e. credit ratings agencies' risk assessments of banks and GDP growth (column 3), we find that neither the sovereign nor bank rating has a significant impact on the sovereign-bank link. However, and as expected, the coefficient of GDP growth is strongly negative and significant. Financial market stress remains associated with significantly higher correlations between the CDS premiums. As a simple means of testing for potential autocorrelation, the lagged dependent variable can be included (column 4). In this specification, the direct sovereign-to-bank and bank-to-sovereign risk transfer channels and VIX remain significant.

If correlations between CDS premiums are tied primarily to implicit guarantees of the sovereign, we expect this effect to be stronger in countries where the government is more likely to finance a bail-out of the banking system, and where banks are larger relative players in the financial system and sovereign debt markets. This could be the case in bank-based financial systems, i.e. those countries where the banking system is more important for aggregate private credit (Gambacorta et al., 2014). Thus, we test the second part of our first hypothesis by interacting our measure of implicit guarantees with the share of bank credit in total credit to the economy. Column (5) of table 3 shows the results of this regression. While the coefficient of the interaction term between exposures and the bank share in credit is positive, it is not statistically significant (t-value of 1.32). The coefficient of implicit liabilities is now much larger, but the negative interaction term implies that this effect is *decreasing* in the share of bank credit. This means that, ceteris paribus, higher implicit contingent liabilities are tied to greater interdependencies primarily in market-based financial systems, and not in highly bank-based systems.¹³ Overall, we can only confirm the first half of our first hypothesis. Higher exposures of banks to sovereigns and a larger banking sector are indeed tied to a higher market perception of sovereign-bank interdependencies, but we cannot confirm that this effect is larger in bank-based financial systems.

The empirical testing of our <u>second hypothesis</u> goes more deeply into the feedback mechanism between banks and sovereigns, as we analyze how direct risk transfer channels, sovereign fundamentals and sovereign credit spreads influence bank credit spreads, and how these links, bank fundamentals and bank credit spreads influence sovereign spreads. We analyze how these effects are manifested in both stressed and tranquil market conditions by interacting the CDS premiums of the other sector with financial market volatility (the VIX).

We start with bank CDS (table 4). First, we find that without controlling for sovereign fundamentals, there is a statistically significant link between the level of bank CDS spreads and the scale of bank exposures to sovereigns (column 1). The measure of implicit contingent liabilities, which could measure the uplift from expectations of sovereign support, is not significant. Second, moving on to the risk measures, we confirm that CDS premiums of banks are not only affected by their own risk characteristics, but also by the riskiness of the home sovereign. We find that the link between a bank's CDS spread and its sovereign exposures seems to be particularly strong when the domestic sovereign has a *higher* credit rating (column 2). This may imply that the value of (implicit)

¹³ Separately, we have found that when splitting the sample into relatively bank-based versus relatively market-based systems, bank exposures are separate in both sub-samples, while implicit contingent liabilities are significant in neither.

sovereign support is higher in countries with a relatively creditworthy sovereign. The sovereign rating, itself, has a significant negative coefficient, with each one notch upgrade being associated with a roughly 29 bp improvement in spreads (column 3). When the lagged sovereign CDS spread is included as an independent variable (column 4), this has a coefficient close to 1. In this specification, sovereign exposures become insignificant, but the implicit liabilities do show up with the expected negative sign, implying sovereign uplift.

Finally, and most importantly for our second hypothesis, we are interested in how the significance of the sovereign CDS and sovereign rating for bank spreads during depends on market volatility. We find that the relation between the level of sovereign CDS and bank CDS is not significantly affected by the VIX (column 5). In other words, while lagged sovereign CDS levels are a (highly) important determinant of bank CDS levels, the sovereign-to-bank risk transfer does not appear to differ between low and high-volatility periods.

Table 4: Estimations of bank CDS with I	xed effects				
	(1)	(2)	(3)	(4)	(5)
		Full s	ample		
Sovereign to Bank risk transfer					
Bank exposures to sovereigns (% total assets)	765.114**	775.413*	-61.1	76.909	26.926
	(289.847)	(400.412)	(199.703)	(137.161)	(152.092)
Exposures to sovereigns * sovereign rating		-133.601**			
		(56.479)			
Bank to Sovereign risk transfer					
Implicit CL (RWA-TE, % of gdp)	-38.868	-24.89	-28.634	-110.814**	-108.698*
	(77.738)	(46.443)	(52.683)	(50.600)	(56.259)
Risk measures					
Sovereign rating		-52.480***	-28.921**	-3.675	
		(10.096)	(11.993)	(8.005)	
Bank rating			-11.771**	-1.075	
			(5.120)	(5.019)	
Macroeconomic controls					
GDP growth			-10.989***	-0.743	
			(2.567)	(1.958)	
Financial market sentiment					
VIX	4.211***	5.520***	4.790***	3.676***	2.545***
	(0.641)	(0.656)	(0.641)	(0.526)	(0.659)
Sovereign sentiment					
Sovereign CDS				0.973***	1.169***
				(0.073)	(0.196)
Sovereign CDS * VIX					-0.007
-					(0.005)
Number of observations	1832	1832	1828	1772	1776
Number of groups	65	65	65	65	65
			50		
R-sq within	0.063	0.264	0.292	0.623	0.598
R-sq between	0.012	0.333	0.475	0.388	0.384
R-sq overall	0.022	0.246	0.337	0.524	0.514

Note: * significance at 90%; ** significance at 95%; *** significance at 99%. Constants are not reported.

For sovereign CDS (table 5), on the other hand, we do find a positive interaction effect, i.e. higher spillovers when the VIX is high. These regressions, which are performed at the country level, again begin with only the direct risk transfer channels (columns 1 and 2). Here, the coefficient for the total banking sector's exposures to sovereigns (now from IMF IFS, at quarterly frequency and on a locational basis) is positive but insignificant. When including the sovereign rating and the interaction of this rating and the sovereign exposures of the banking sector, we find a significantly negative coefficient for the sovereign rating, as expected, but no effect of the rating on the significance of sovereign exposures (column 2). The implicit liabilities measure, which is a sum for all of the banks in sample and should be associated with higher sovereign CDS spreads, is not significant. When the bank rating is added (columns 3 and 4), it has a significant negative sign, with a one-notch upgrade of any major bank being associated with a decline in the sovereign CDS spread of 18 bp. Note that the coefficient of the bank rating is higher than that of the sovereign rating, with the sovereign rating even becoming insignificant after adding the lagged average CDS spread of the banks in sample (weighted by total assets) as an explanatory variable. As expected, higher output growth is also linked to lower sovereign CDS spreads in the respective quarter.

Finally, and again important for our second hypothesis, we consider the role of market volatility in the relation between (lagged) banking CDS spreads and the sovereign CDS (column 5). Here, we find that market volatility is associated with a significantly stronger link between the banking CDS and sovereign CDS. This shows that the riskiness of the banking sector matters more for market perceptions of sovereign creditworthiness during stressed market conditions.

	(1)	(2)	(3)	(4)	(5)
		Full s	ample		
Sovereign to Bank					
Bank exposures to sovereigns (% total assets)	404.845	-77.253	-487.335	-519.300	52.346
	(453.884)	(663.953)	(570.003)	(550.387)	(262.582)
Exposures to sovereigns * sovereign rating		-86.157			
		(57.301)			
Bank to Sovereign					
Implicit CL (RWA-TE, % of gdp)	22.291	41.286	31.84	62.074	68.754
	(30.332)	(30.648)	(24.093)	(44.761)	(47.381)
Risk measures					
Sovereign rating		-43.446***	-16.143*	-3.954	
		(14.253)	(9.194)	(5.837)	
Bank rating			-17.944***	-15.973***	
Ũ			(3.752)	(4.851)	
Macroeconomic controls					
GDP growth			-5.212***	-2.950*	
			(1.773)	(1.575)	
Financial market sentiment					
VIX	2.125***	2.817***	3.316***	2.642***	-2.060*
	(0.664)	(0.777)	(0.750)	(0.775)	(1.079)
Banking sentiment					
Banking CDS				0.165	-0.044
				(0.098)	(0.075)
Banking CDS * VIX					0 014***
					(0.003)
Number of observations	737	737	737	732	732
Number of groups	23	23	23	732	23
rumoer of groups	23	23	23	23	23
R-sq within	0.037	0.242	0.267	0.354	0.390

Table 5: Estimations of sovereign CDS with fixed effects

Note: * significance at 90%; ** significance at 95%; *** significance at 99%. Constants are not reported.

0.107

0.051

0.646

0.376

0.750

0.421

0.244 0.254 0.134

0.215

4.2 Robustness checks

R-sq between

R-sq overall

The above results have been subjected to a number of tests in which we alter the model specifications and the way the variables are constructed. These tests generally support the key results of our baseline regressions, though statistical significance (particularly of implicit contingent liabilities) sometimes changes. The results of several tests on the correlations are presented in table 6.

One possible criticism of our baseline regression is that the VIX, as a global variable, may not be equally appropriate for each country in the sample. One alternative measure that has recently become available for major economies is the economic policy uncertainty index of Baker, Bloom and Davis (2015). This index, based on the frequency of key words regarding economic policy uncertainty in international newspapers, is positively correlated with the VIX, but is more sensitive to national specificities. The first column of table 6 shows the results when using the economic policy uncertainty index. The sample size is lower due to lower country coverage. Bank exposures remain significant, while implicit contingent liabilities are insignificant.

	Bank level			Country level	
	(1)	(2)	(3)	(4)	(5)
	Baker / Bloom	Euro area	Non-Euro	Exposures EU	Exposures EU
	/ Davis index	countries	area		
Sovereign to Bank					
Bank exposures to sovereigns (% total assets)	1.046*	1.178***	0.466***	4.772**	-13.243
	(0.534)	(0.306)	(0.170)	(1.666)	(9.249)
Exposures to sovereigns * bank share					26.995*
					(13.925)
Bank to Sovereign	0.157	0 100***	0.075	0.0/2**	0.142
Implicit CL (RWA-1E, % of GDP)	0.157	0.123***	0.075	0.063**	-0.143
	(0.231)	(0.040)	(0.055)	(0.022)	(0.097)
Implicit CL * bank share					0.392*
					(0.194)
Risk measures					
Sovereign rating	-0.004	-0.008*	0.033***	0.018	
	(0.012)	(0.005)	(0.008)	(0.016)	
Bank rating	0.02	0.009*	0.015**	-0.019	
	(0.014)	(0.005)	(0.007)	(0.018)	
Macroeconomic controls					
GDP growth	-0.018***	-0.003	-0.002	-0.017***	
	(0.005)	(0.002)	(0.002)	(0.005)	
Bank share					-2.007*
					(0.926)
Financial market sentiment					
Economic policy uncertainty	0.091***				
	(0.016)				
VIX		0.004***	0.004***	0.008***	0.008***
		(0.001)	(0.001)	(0.001)	(0.001)
Autocorrelation					
Spearman (lag1)		0.574***	0.449***		
		(0.044)	(0.043)		
Number of observations	976	691	858	389	389
Number of groups	37	26	38	11	11
R-sq within	0.114	0.472	0.355	0.240	0.207
R-sq between	0.017	0.447	0.102	0.125	0.036
R-sq overall	0.003	0.471	0.017	0.177	0.063

Table of filler name of commanding of corrections between bank and boyer eight of b

Note: * significance at 90%; ** significance at 95%; *** significance at 99%.

A second potential criticism is that the effects may be different across jurisdictions, and in particular in the members of a currency union – particularly the euro area – than in countries with their own currency (De Grauwe and Ji, 2013; Van Loon and De Haan, 2015). When dividing our sample (columns 2 and 3) we can confirm that both bank exposures and contingent liabilities have a stronger association with CDS interdependency for banks inside the euro area. Sovereign exposures are also highly significant for non-euro area banks, while contingent liabilities are not.

Moreover, for euro area banks, higher sovereign ratings are tied to a lower interdependency, while higher bank ratings are associated with a tighter sovereign-bank link. These results warrant further investigation, ideally taking into account further institutional characteristics, such as regulatory regimes for sovereign debt and state aid rules in the various countries.

Where available, we use information on bank exposures to the domestic sovereign rather than the broader class of sovereign exposures (including Sweden, but with the exception of the UK). Although this is a more precise test of our hypotheses, the information is only available for EU countries. Moreover, we define the bank-sovereign correlations at the country level, i.e. the correlations between the domestic sovereign and an index of the banks in sample from that country, weighted by total assets (columns 4 and 5). Here, we find that bank exposures to the domestic sovereign (from ECB) and the level of implicit liabilities are highly significant. When we interact these variables with the bank share in total credit, we find that both bank exposures to the domestic sovereign and the level of implicit liabilities are more strongly related to the CDS correlation if the share of bank credit is higher. While this exercise has a number of caveats (including fewer observations and aggregation of important bank-specific indicators), these results are fully in line with the first hypothesis. Again, this suggests there may be potential for further investigating the differences between bank-based and market-based economies, and for the specific characteristics of the euro area.

Finally, additional checks have confirmed that the results are robust to using a simple Pearson's correlation as the dependent variable, and to alternative definitions of implicit contingent liabilities, including the size of the banking sector, leverage ratios or the largest banks in percent of GDP. The sign and significance of coefficients are comparable to the baseline regressions. These results are omitted for brevity but are available upon request.

4.3 Extensions

In general, we would expect to find positive correlations between the sovereign CDS and that of other domestic counterparties, both financial institutions and non-financial corporates. Both sovereigns and private borrowers are subject to shocks in the domestic economy, and the riskiness of the sovereign is an important determinant of the credit rating of domestic borrowers. However, implicit in our analysis is the idea that the relationship between banks and sovereigns is "special." Banks are typically large holders of sovereign debt, whereas the government is more inclined to use large amounts of financial support for the banking sector than for other sectors in the economy. (By way of illustration: governments bail out banks, like ING, but are much less likely to bail out large corporates, like Royal Dutch Shell). So even though the risk premiums of the sovereign and a wide range of domestic borrowers may be correlated, we would expect this to be the result of the (indirect) links through dependence on the real economy and, to a lesser extent, financial markets (channels 3 and 4 in figure 1).



Figure 4: The sovereign-bank nexus versus the "sovereign-corporate nexus"

Note: Bars show the average correlation for country groups in sample. The correlation between corporate and sovereign CDS spreads uses the same calculation method as for the sovereign bank-correlation: a Spearman rank correlation calculated with daily data, based on the data within the respective quarter.

When we compare the level of correlation of sovereign and bank CDS premiums to that of sovereign and corporate CDS premiums, we find no large differences on average. But when we divide the sample into countries with a bank based versus a market based financial system, we find that the difference between the two sub-samples is substantially larger for the sovereign-bank correlation (figure 4). Moreover, when rerunning the above regressions at the country level with the correlation between sovereign CDS and average corporate CDS premiums as the dependent variable (not reported), we find a stronger effect for changes in GDP growth and a weaker effect for our direct risk transfer channels than for our baseline regressions with bank CDS premiums. This illustrates that the existence of direct feedback mechanisms is especially important for sovereigns and banks, and is again an invitation for further analysis.

5. Conclusions

In this study, we have investigated four types of channels which may impact sovereign-bank interdependencies across countries and time. With our empirical analysis, we can confirm that the two direct channels of risk transfer – through sovereign debt holdings of banks, which expose banks to sovereign risk shocks, and risk-weighted assets minus total equity relative to GDP, which measure implicit contingent liabilities to banks – are linked to higher correlations between sovereign and bank CDS. Thus, the market perception of interdependency between sovereign and banking credit risk does respond to fundamental factors. These relationships are present when controlling for changes in sovereign and bank ratings, macroeconomic factors and market volatility. Yet somewhat surprisingly, we do not find these effects to be stronger in bank-based than in market-

based financial systems. We can thus confirm the importance of direct risk transfer channels, but not the expectation of a stronger effect in bank-based systems, at least when the analysis is done at the level of individual banks.

From a systemic risk perspective, such correlations are relevant primarily if they lead to higher financing costs for banks or sovereigns, particularly in periods of stress. Our second hypothesis, which relates to the impact of sovereign (bank) credit premiums on the level of bank (sovereign) credit premiums, offers a more complete picture. When volatility is high, markets are more likely to consider the riskiness of banks as relevant for sovereign risk and to charge higher financing costs. Banks with higher sovereign exposures may also pay higher spreads, but the impact of sovereign premiums does not appear to be higher in periods of high volatility. This appears to underscore the importance of sovereign-to-bank risk transfer.

Overall, our results show that market data across countries and time may hold important clues to how systemic risk is manifested. Especially CDS markets hold great potential for empirical analysis. It is perhaps tempting during bouts of high market volatility to claim that markets are overreacting and that credit spreads in the financial markets do not correspond to fundamentals. While this may or may not be true and objectively measurable, it has to be recognized that market prices do matter for the liquidity and solvency of both sovereigns and banks, and do have real effects on overall macroeconomic outcomes. From a systemic risk perspective, it is thus important to consider how policy choices – such as the preferential treatment of sovereign debt in prudential regulation, large and undercapitalized banking sectors, or expectations of state support to banks – may influence the probability and severity of financial shocks and "twin crises." Policies such as the zero risk weighting for sovereign exposures in Basel III, the lack of large exposure limits and numerous other exemptions for banks' holdings of sovereign debt seem particularly relevant in this context. Overall, our results at the least provide grounds for caution when considering the retention of such policy measures which strengthen sovereign-bank interdependencies.

Annex: Countries and banks in sample

Table A1 shows the list of countries and banks included in this study. Our sample of 23 countries is determined largely by the availability of CDS data, i.e. the presence of an active CDS market, and reliable macroeconomic and bank balance sheet data. Our sample includes 15 countries categorized by Laeven and Valencia (2013) as having systemic banking crisis over 2007-2010, and 8 countries without.¹⁴ Moreover, we have 13 EU members and 5 emerging economies as classified by the IMF. Banks are selected based on the existence of an active CDS market, with prices that change on at least 5 trading days within a single quarter. In general, these are the most systemically important institutions within their respective jurisdictions.

Country	Banks
Australia*	Commonwealth Bank of Australia, Westpac Bank, ANZ Bank
Austria*	Raiffeisen Zentralbank, BAWAG, Erste Group, UniCredit Bank Austria
Belgium*	KBC, Dexia
Brazil	Banco do Brasil, Brasil Bradesco, BNDES
China	Bank of China
Denmark	Danske Bank
France*	BNP Paribas, Société Générale, Crédit Agricole
Germany	Commerzbank, Deutsche Bank
India	Bank of India, Exim Bank of India, ICICI Bank, IDBI Bank, State Bank of India
Ireland*	Bank of Ireland, Allied Irish
Italy	Intesa Sanpaolo, UniCredit, Mediobanca
Japan*	Mitsubishi, Sumitomo Mitsui, Mizuho
Netherlands*	ABN Amro Bank, ING Bank, Rabobank
Norway*	DNB Norway Bank
Portugal	Banco Commercial Portugues, Banco Espirito Santo, Caixa Geral
Republic of Korea	Industrial Bank of Korea, Kookmin Bank, Shinhan Bank, Woori Bank
Russian Federation	Alfa Bank, Bank of Moscow, VTB Bank, Sberbank
Spain	Santander, BBVA, Sabadell
Sweden*	SEB, Svenska Handelsbanken, Swedbank, Nordea
Switzerland	UBS, Credit Suisse
Turkey	Akbank, Türkiye Iş Bankası
United Kingdom*	Barclays, Lloyds, HSBC
United States*	Wells Fargo, Citigroup, JPMorgan Chase

Table A1: Countries and banks used for estimations

* Countries classified as relatively market-based economies based on a ratio of bank credit in total private credit that is below the sample median.

¹⁴ The countries without recent banking crises are Japan, Turkey, China, Brazil, India, Korea, Norway and Australia.

References

- Acharya, Viral, Itamar Drechsler and Philipp Schnabl (2014). A Pyrrhic victory? Bank bailouts and sovereign credit risk. *The Journal of Finance* 69(6): 2689-2739.
- Acharya, Viral and Sascha Steffen (2015). The "Greatest" Carry Trade Ever? Understanding Eurozone Bank Risks. *Journal of Financial Economics* 115(2): 215-236.
- Alter, Adrian and Yves Schüler (2012). Credit spread interdependencies of European states and banks during the financial crisis. *Journal of Banking and Finance* 36: 3444-3468.
- Angelini, Paolo, Giuseppe Grande, and Fabio Panetta (2014). The negative feedback loop between banks and sovereigns. Bank of Italy Occasional Paper No. 213.
- Asonuma, Tamon, Said Bakhache, and Heiko Hesse (2015). Is Banks' Home Bias Good or Bad for Public Debt Sustainability? IMF Working Paper WP/15/44.
- Balteanu, Irina and Aitor Erce (2014). Banking crises and sovereign defaults in emerging markets: exploring the links. Banco de España Working Paper No. 1414.
- Baur, Dirk and Niels Schulze (2005). Coexceedances in financial markets a quantile regression analysis of contagion. *Emerging Markets Review* 6(1): 21-43.
- Bénassy-Quéré, Agnes and Guillaume Roussellet (2014). Fiscal sustainability in the presence of systemic banks: the case of EU countries. *International Tax and Public Finance* 21(3): 436-467.
- BIS (2011). The impact of sovereign credit risk on bank funding conditions. Bank for International Settlements, Basel.
- Blundell-Wignall, Adrian and Patrick Slovik (2011). A Market Perspective on the European Sovereign Debt and Banking Crisis. *OECD Journal* 2010 (2).
- Bolton, Patrick and Olivier Jeanne (2011). Sovereign Default Risk and Bank Fragility in Financially Integrated Economies. *IMF Economic Review* 59(2): 162-194.
- Borensztein, and Ugo Panizza (2009). The Costs of Sovereign Default. *IMF Staff Papers* 56(4): 683-741.
- Bottero, Margherita, Simone Lenzu and Filippo Mezzanotti (2015). Sovereign debt exposure and the bank lending channel: impact on credit supply and the real economy. Banca d'Italia Working Paper No. 1032.
- Breckenfelder, H.-Johannes and Bernd Schwaab (2015). The bank-sovereign nexus across borders. *mimeo*.
- Broner, Fernando, Aitor Erce, Alberto Martin and Jaume Ventura (2014). Sovereign Debt Markets in Turbulent Times: Creditor Discrimination and Crowding-Out. *Journal of Monetary Economics* 61: 1-172.
- Castro, Christian and Javier Mencía (2014). Sovereign risk and financial stability. *Revista de Estabilidad Financiera* 26: 73-107.
- Collin-Dufresne, Pierre, Robert S. Goldstein, and J. Spencer Martin (2001). The determinants of credit spread changes. *Journal of Finance* 56: 2177-2207.
- Cooper, Russell and Kalin Nikolov (2015). Government Debt and Banking Fragility: The Spreading of Strategic Uncertainty. NBER Working Paper No. 19278.
- De Grauwe, Paul and Yuemei Ji (2013). Self-fulfilling Crises in the Eurozone: An Empirical Test. *Journal of International Money and Finance* 34: 15-36.
- De Haan, Leo, Jeroen Hessel and Jan Willem van den End (2013). Are European sovereign bonds fairly priced? The role of modeling uncertainty. DNB Working Paper No. 399.
- Dieckmann, S. and T. Plank (2011). Default risk of advanced economies: an empirical analysis of credit default swaps during the financial crisis. *Review of Finance* 0: 1-32.
- ESRB (2015). ESRB report on the regulatory treatment of sovereign exposures. European Systemic Risk Board, Frankfurt am Main.
- Forbes, Kristin and Roberto Rigobon (2002). No Contagion, Only Interdependence: Measuring

Stock Market Comovements. Journal of Finance 57(5): 2223-2261.

- Gambacorta, Leonardo, Jing Yang and Kostas Tsatsaronis (2014). Financial structure and growth. BIS Quarterly Review, March.
- Goldsmith, Raymond W. (1969). *Financial Structure and Development*. New Haven and London: Yale University Press.
- Gray, D.F., R.C. Merton and Z. Bodie (2008). New framework for measuring and managing macrofinancial risk and financial stability. NBER Working Papers No. 13607.
- Gray, D.F., and A.A. Jobst (2013). Systemic Contingent Claims Analysis; Estimating Market-Implied Systemic Risk. IMF Working Paper No. 13/54.
- Gros, D. (2013). Banking Union with a Sovereign Virus: The self-serving regulatory treatment of sovereign debt in the euro area. CEPS Working Paper No 289.
- Haugh, D., Ollivaud, P., Turner, D. (2009). What Drives Sovereign Risk Premiums?: An Analysis of Recent Evidence from the Euro Area. OECD Economics Department Working Paper No. 718.
- IMF (2010). Global Financial Stability Report Sovereign, Funding and Systemic Liquidity. Washington, DC.
- Laeven, Luc and Fabián Valencia (2013). Systemic banking crises database. *IMF Economic Review* 61(2): 225-270.
- Levin, Andrew, Chien-Fu Lin and Chia-Shang James Chu (2002). Unit root tests in panel data: asymptotic and finite-sample properties. *Journal of Econometrics*, 108(1): 1-24.
- Levine, Ross (2002). Bank-Based or Market-Based Financial Systems: Which Is Better? *Journal* of Financial Intermediation 11(4): 398-428.
- Loon, Eric van and Jakob de Haan (2015). Location of banks and their credit ratings. *The Journal* of *Risk Finance* 16(3): 220-232.
- Merler, Silvia and Jean Pisani-Ferry (2012). Hazardous tango: sovereign-bank interdependence and financial stability in the euro area. Banque de France Financial Stability Review No. 16.
- Mink, Mark and Jakob de Haan (2013). Contagion during the Greek sovereign debt crisis. *Journal of International Money and Finance* 34: 102-113.
- Ongena, Steven, Alexander Popov and Neeltje van Horen (2016). The invisible hand of the government: "Moral suasion" during the European sovereign debt crisis. DNB Working Paper No. 505.
- Popov, Alexander and Neeltje van Horen (2012). The impact of sovereign debt exposure on bank lending: Evidence from the European debt crisis. DNB Working Paper No. 382.
- Reinhart, Carmen and Kenneth Rogoff (2009). *This Time Is Different: Eight Centuries of Financial Folly*. Princeton, New Jersey: Princeton University Press.
- Reinhart, Carmen and Kenneth Rogoff (2011). From Financial Crash to Debt Crisis. *American Economic Review*, 101(5): 1676-1706.
- Zettelmayer, Jeromin, Christoph Trebesch and Mitu Gulati (2013). The Greek Debt Restructuring: An Autopsy. Peterson Institute for International Economics Working Paper No. 2013-13-8.

Previous DNB Working Papers in 2016

- No. 493 Jacob Bikker, Dirk Gerritsen and Steffie Schwillens, Competing for savings: how important is creditworthiness during the crisis?
- No. 494 Jon Danielsson and Chen Zhou, Why risk is so hard to measure
- No. 495 **Gabriele Galati, Irma Hindrayanto, Siem Jan Koopman and Marente Vlekke**, Measuring financial cycles with a model-based filter: Empirical evidence for the United States and the euro area
- No. 496 Dimitris Christelis, Dimitris Georgarakos, Tullio Jappelli and Maarten van Rooij, Consumption uncertainty and precautionary saving
- No. 497 Marco Hoeberichts and Ad Stokman, Price level convergence within the euro area: How Europe caught up with the US and lost terrain again
- No. 498 Janko Cizel, Jon Frost, Aerdt Houben and Peter Wierts, Effective macroprudential policy: Crosssector substitution from price and quantity measures
- No. 499 **Frank van der Horst, Martina Eschelbach, Susann Sieber and Jelle Miedema**, Does banknote quality affect counterfeit detection? Experimental evidence from Germany and the Netherlands
- No. 500 Jochen Mierau and Mark Mink, A descriptive model of banking and aggregate demand
- No. 501 **Clemens Bonner, Daniel Streitz and Michael Wedow**, On the differential impact of securitization on bank lending during the financial crisis
- No. 502 **Mijntje Lückerath-Rovers and Margriet Stavast-Groothuis**, The changing composition of the supervisory boards of the eight largest banks and insurers during 2008-2014 and the impact of the "4+4 suitability screenings"
- No. 503 Dirk Broeders, Damiaan Chen, Peter Minderhoud and Willem Schudel, Pension funds' herding
- No. 504 **Ronald Heijmans, Richard Heuver and Zion Gorgi**, How to monitor the exit from the Eurosystem's unconventional monetary policy: Is EONIA dead and gone?
- No. 505 Steven Ongena, Alexander Popov and Neeltje Van Horen, The invisible hand of the government: "Moral suasion" during the European sovereign debt crisis
- No. 506 Wändi Bruine de Bruin, Wilbert van der Klaauw, Maarten van Rooij, Federica Teppa and Klaas de Vos, Measuring expectations of inflation: Effects of survey mode, wording, and opportunities to revise
- No. 507 Jos Jansen and Jasper de Winter, Improving model-based near-term GDP forecasts by subjective forecasts: A real-time exercise for the G7 countries
- No. 508 Malka de Castro Campos and Federica Teppa, Individual inflation expectations in a declininginflation environment: Evidence from survey data
- No. 509 Gabriele Galati, Zion Gorgi, Richhild Moessner and Chen Zhou, Deflation risk in the euro area and central bank credibility
- No. 510 Christiaan Pattipeilohy, A comparative analysis of developments in central bank balance sheet composition
- No. 511 **Guido Ascari, Andrea Colciago and Lorenza Rossi**, Determinacy analysis in high order dynamic systems: The case of nominal rigidities and limited asset market participation
- No. 512 David-Jan Jansen and Richhild Moessner, Communicating dissent on monetary policy: Evidence from central bank minutes
- No. 513 Leo de Haan and Maarten van Oordt, Timing of banks' loan loss provisioning during the crisis
- No. 514 Cenkhan Sahin, Macroeconomic effects of mortgage interest deduction
- No. 515 Karsten Staehr and Robert Vermeulen, How competitiveness shocks affect macroeconomic performance across euro area countries
- No. 516 Leo de Haan and Jan Willem van den End, The signalling content of asset prices for inflation: Implications for Quantitative Easing
- No. 517 **Daniël Vullings**, Contingent convertible bonds with floating coupon payments: fixing the equilibrium problem.
- No. 518 Sebastiaan Pool, Credit Defaults, Bank Lending and the Real Economy
- No. 519 **David-Jan Jansen and Nicole Jonker**, Fuel tourism in Dutch border regions: are only salient price differentials relevant?
- No. 520 Jon Frost, Jakob de Haan and Neeltje van Horen, International banking and cross-border effects of regulation: lessons from the Netherlands
- No. 521 Wilko Bolt and Maarten van Oordt, On the value of virtual currencies
- No. 522 David-Jan Jansen, Housing and mortgage dynamics: evidence from household surveys
- No. 523 Michelle Bongard, Gabriele Galati, Richhild Moessner and William Nelson, Connecting the dots: market reactions to forecasts of policy rates and forward guidance provided by the Fed
- No. 524 **Dennis Bonam and Bart Hobijn**, Generalized stability of monetary unions under regime switching in monetary and fiscal policies

Previous DNB Working Papers in 2016 (continued)

- No. 525 Alan Blinder, Michael Ehrmann, Jakob de Haan and David-Jan Jansen, Necessity as the mother of invention: monetary policy after the crisis
- No. 526 **Raymond Chaudron**, Bank profitability and risk taking in a prolonged environment of low interest rates: a study of interest rate risk in the banking book of Dutch banks
- No. 527 Steven Poelhekke, Financial globalization and foreign direct investment
- No. 528 Marco van der Leij, Daan in 't Veld and Cars Hommes, The formation of a core-periphery structure in heterogeneous financial networks
- No. 529 **Yimin Xu and Jakob de Haan**, Does the Fed's unconventional monetary policy weaken the link between the financial and the real sector?
- No. 530 Jakob de Haan and Jan-Egbert Sturm, Finance and income inequality: A review and new evidence
- No. 531 Martijn Boermans and Robert Vermeulen, International investment positions revisited: Investor heterogeneity and individual security characteristics
- No. 532 Carin van der Cruijsen and Frank van der Horst, Payment behaviour: the role of socio-psychological factors
- No. 533 Ralph De Haas and Steven Poelhekke, Mining matters: Natural resource extraction and local business constraints
- No. 534 Mark Mink, Aggregate liquidity and banking sector fragility
- No. 535 **Carin van der Cruijsen and Nicole Jonker**, Pension profile preferences: the influence of trust and expected expenses
- No. 536 Niels Gilbert and Sebastiaan Pool, Sectoral allocation and macroeconomic imbalances in EMU
- No. 537 Dimitris Christelis, Dimitris Georgarakos, Tullio Jappelli and Maarten van Rooij, Trust in the central bank and inflation expectations
- No. 538 Maarten van Rooij en Jakob de Haan, Will helicopter money be spent? New evidence
- No. 539 Jakob de Haan and Sylvester Eijffinger, The politics of central bank independence
- No. 540 Mark Mink, Jan Jacobs and Jakob de Haan, Euro area imbalances

DeNederlandscheBank

De Nederlandsche Bank N.V. Postbus 98, 1000 AB Amsterdam 020 524 91 11 dnb.nl